Author's response to reviews

Title: Evaluation on Intervention Effects of Work-related Musculoskeletal Disorders among School Teachers

Authors:

Jian Shuai (jamedannyhandsome@yahoo.com)
Pengying Yue (g_daisy@126.com)
Liping Li (lpli@stu.edu.cn)
Fengying Liu (liufengying85@163.com)
Sheng Wang (shengw@bjmu.edu.cn)

Version: 2
Date: 17 June 2014

Author's response to reviews: see over
Dear Editors,

We would like to submit the enclosed manuscript entitled “Evaluation on Intervention Effects of Work-related Musculoskeletal Disorders among School Teachers”, which we wish to be considered for publication in “BMC Public Health”. No conflict of interest exits in the submission of this manuscript, and manuscript is approved by all authors for publication. I would like to declare on behalf of my co-authors that the work described was original research that has not been published previously, and not under consideration for publication elsewhere, in whole or in part. All the authors listed have approved the manuscript that is enclosed.

Musculoskeletal disorders represent one of the most common and most expensive occupational health problems in both developed and developing countries. While there is a large published literature in relation to teachers and WMSDs going back 10 years or more, little has been published on WMSDs of teachers, including intervention studies, especially in China. Our study had compare the awareness, attitude, behavior and self-reported work-related musculoskeletal disorders prevalence among teachers between pre-intervention, 6 months and 12 months post-intervention. To explore the effect of short-term and long-term intervention among teachers. We demonstrate the effectiveness of multifaceted ergonomic intervention designed to reduce musculoskeletal symptoms and improve work efficiency in teachers. I hope this paper is suitable for “BMC Public Health”.

We deeply appreciate your consideration of our manuscript, and we look forward to receiving comments from the reviewers. If you have any queries, please don’t hesitate to contact me at the address below.

Jian Shuai

Injury Prevention Research Center, Shantou University Medical College, No.22 Xin Ling Road, Shantou, Guangdong Province, China. 515041

Email: jamedannyhandsome@yahoo.com

Fax: 0086 754 88550372

Mobile telephone: 13790834210

Thank you and best regards.

Yours sincerely

We revise our manuscript to include:

1. line numbers.

2. page numbers.

3. Adjusted the line spacing of references.

Caused any inconvenience to you, hope your understanding. Thank you!